Author's response to reviews

Title: Status and influencing factors of health behavioral knowledge among the elderly in China : A community-based cross-sectional study

Authors:

Zhiqin Yin (yzq@wzmc.edu.cn)
Guiling Geng (gengguiling@yahoo.com.cn)
Xuefen Lan (lanxuefen2008@163.com)
Liming Zhang (920425722@qq.com)
Surong Wang (wsr@ntu.edu.cn)
Yuantong Zang (42352887@qq.com)
Meidi Peng (pengmeidi@ntu.edu.cn)

Version: 2 Date: 13 January 2013

Author's response to reviews:

Dear Jimmar Dizon#

We are pleased to make the amendment of you and the manuscript (Manuscript number 3932161418739521, Title: Status and influencing factors of health behavioral knowledge among the elderly in China) has also been extensively revised according to the comments and was resubmitted online.

Revision #1: We inserted “A community-based cross-sectional study” in the title and the email addresses of all authors were at the title page, which marked in red in the paper.

Revision #2: We moved the Abstract on the next page following the title page.

Revision #3: We have studied your advise carefully and have made revision of Abstract, which marked in yellow in the paper.

Revision #4: We inserted Competing interests, Authors’ contributions and Acknowledgements after the Conclusions, which marked in red.

Revision #5: We deleted a paragraph in discuss section, which marked in blue.

Revision #6: We insert a sentence “but the elderly people and their children are similar in their knowledge of health, while some are good, others are not.” in discuss section, which marked in red.

Revision #7: We change References[36] into [35], [37] into [36], [38] into [37].

Best Wishes,
Sincerely yours,
Zhiqin Yin